Author's response to reviews

Title: Patient Safety skills in Primary Care: a national survey of GP educators

Authors:

Maria Ahmed (maria.k.ahmed@gmail.com)
Sonal Arora (sonal.arora06@imperial.ac.uk)
John McKay (John.McKay@nes.scot.nhs.uk)
Susannah Long (s.long@imperial.ac.uk)
Charles Vincent (charles.vincent@psy.ox.ac.uk)
Moya Kelly (Moya.Kelly@nes.scot.nhs.uk)
Nick Sevdalis (n.sevdalis@imperial.ac.uk)
Paul Bowie (Paul.Bowie@nes.scot.nhs.uk)

Version: 2
Date: 27 May 2014

Author's response to reviews:

Dear Editorial team

I have inserted line numbers as requested

Best Wishes

Dr Maria Ahmed